

Fuzzy C-means for english sentiment classification in a distributed system

Vo Ngoc Phu
^{1,2} $\bigcirc \cdot$ Nguyen Duy Dat
^3 \cdot Vo Thi Ngoc Tran
4 \cdot Vo Thi Ngoc Chau
^5 \cdot Tuan A. Nguyen
^6

Published online: 5 November 2016 © Springer Science+Business Media New York 2016

Abstract Sentiment classification plays a significant role in everyday life, in political activities, in activities relating to commodity production, and commercial activities.

☑ Vo Ngoc Phu vongocphu@tdt.edu.vn

> Nguyen Duy Dat duydatspk@gmail.com

> Vo Thi Ngoc Tran vtntran@hcmut.edu.vn

Vo Thi Ngoc Chau chauvtn@cse.hcmut.edu.vn; chauvtn@hcmut.edu.vn; chauvtn2003@gmail.com

Tuan A. Nguyen tuanna@uit.edu.vn

- ¹ Division of Computational Mathematics and Engineering, Institute for Computational Science, Ton Duc Thang University, Ho Chi Minh City, Vietnam
- ² Faculty of Information Technology, Ton Duc Thang University, Ho Chi Minh City, Vietnam
- ³ Faculty of Information Technology, Ly Tu Trong Technical College, Ho Chi Minh City, Vietnam
- ⁴ School of Industrial Management (SIM), Ho Chi Minh City University of Technology, Vietnam National University, Ho Chi Minh City, Vietnam
- ⁵ Computer Science & Engineering (CSE), Ho Chi Minh City University of Technology, Vietnam National University, Ho Chi Minh City, Vietnam
- ⁶ Faculty of Computer Networks and Communications, University of Information Technology, Vietnam National University of Hochiminh City, Linh Trung Ward, Thu Duc District, Ho Chi Minh City, Vietnam

Finding a solution for the accurate and timely classification of emotions is a challenging task. In this research, we propose a new model for big data sentiment classification in the parallel network environment. Our proposed model uses the Fuzzy C-Means (FCM) method for English sentiment classification with Hadoop MAP (M) /REDUCE (R) in Cloudera. Cloudera is a parallel network environment. Our proposed model can classify the sentiments of millions of English documents in the parallel network environment. We tested our model using the testing data set (which comprised 25,000 English reviews, 12,500 being positive and 12,500 negative) and achieved 60.2 % accuracy. Our English training data set has 60,000 English sentences, comprising 30,000 positive English sentences and 30,000 negative English sentences.

Keywords Sentiment classification · English sentiment classification · Opinion mining · English document opinion mining · Fuzzy C-Means · FCM · Cloudera · Parallel environment · Parallel network · Parallel network environment · Distributed system

1 Introduction

Sentiment classification plays a significant role in everyday life, in political activities, in activities relating to commodity production, and commercial activities. Finding a solution for the accurate and timely classification of emotion is a challenging task.

Data clustering is the process of putting objects into classes where the objects are similar. A cluster is a set of data objects which are similar in scope, but are not similar to objects in other clusters. Number of data clusters are clustered which can be identified firstly following experience or can be automatically identified in the clustering method.

This technique clusters a set of n data object vectors $X = \{x1,x2,...,xn\} \subset Rs$ into c fuzzy clusters based on calculating the minimum objective function to measure the quality of clustering and find the cluster centers in each cluster to minimize the cost measurement function. A fuzzy set is a set in which each x basic member is assigned a $\mu(\xi)$ real value in [0, 1] to display the dependency measure of this member in the set. When the dependency measure equals 0, the basic member does not belong to the set; but if the dependency measure equals 1, the basic member belongs to the set completely. Therefore, a fuzzy set is a set of $(x, \mu(x))$ pairs.

Fuzzy C-Means (FCM) is also a method of clustering which allows one element to belong to two or more clusters. It is often used to cluster the data but seldom used for data classification.

We suggest many basic principles of our model to classify the opinions (positive, negative, neutral) expressed in the English documents in the English testing data set, based on the large number of English sentences in the English training data set. The principle underpinning our proposed model, which uses clustering techniques to classify the semantics of the English documents are as follows: assuming that an English document contains n English sentences, the English document has a positive polarity if the number of English sentences clustered into the 30,000 positive English sentences of the training data set is greater than the number of English sentences clustered into the 30,000 negative English sentences of the training data set. Conversely, the English document has a negative polarity if the number of English sentences clustered into the 30,000 positive English sentences of the training data set is less than the number of English sentences clustered into the 30,000 negative English sentences of the training data set. Finally, the English document has a neutral polarity if the number of English sentences clustered into the 30000 positive English sentences of the training data set is equal to the number of English sentences clustered into the 30,000 negative English sentences of the training data set.

Based on these principles, we implement our proposed model in the Cloudera parallel network environment. Our model uses FCM combined with Hadoop Map (M)/Reduce (R) to classify the sentiments (positive, negative, neutral) of one English document in the English testing data set into either positive polarity, negative polarity or neutral polarity in the Cloudera parallel network environment.

To implement this study, we use the basis Fuzzy C-Means algorithm (the core basis of the Fuzzy C-Means algorithm)

presented in [8–24]. There are also many studies which use the FCM in semantic classification (opinion mining, sentiment analysis) but there is not much work which uses FCM for sentiment analysis with the aforementioned principles of our proposed model.

FCM is a clustering technique in the data mining field and it has been applied in the natural language processing field where we have had many difficulties and it has taken a long time to implement this research. There are many advantages of FCM, such as: it is unsupervised, it always converges; it provides membership values which are useful for interpretation; it is flexible with respect to the distance used; and if some of the membership values are known, this can be incorporated into the numerical optimization. There are several disadvantages of FCM as follows: long computational time; sensitivity to the initial guess (speed, local minima); and sensitivity to noise - one expects low (or even no) membership degree for outliers (noisy points).

In addition, based on the work related to FCM and the sentiment analysis of big data in [8–24], there are not studies which use FCM for big data in sentiment classification. We use FCM in our model opinion mining in big data, although our English data set in this work is a small English testing data set with 25,000 English document in each testing data set.

In addition, based on many works related to FCM in the parallel system (or FCM in the distributed system) in [25–27], many studies relate to parallel systems or distributed systems, in [28–42], FCM used research for sentiment classification in [43–50], and many studies in the world, there is not any study related to FCM for sentiment classification in parallel system but our model uses FMC for semantic analysis in the distributed system.

Many studies, such as [2-56], use Hadoop Map (M)/Reduce (R), and Cloudera; Vector Space Models (VSM); FCM; FCM in parallel systems (distributed systems)sentiment classification and big data. However, to the best of our knowledge, no studies use all of them. Our proposed model uses all of these.

Finally, we build many FCM-related algorithms in our new model based on the basic FCM in the Cloudera distributed system with Hadoop Map (M) /Reduce (R) and these algorithms have not been used in any other study.

This study comprises six sections: Section 1 is the introduction; Section 2 discusses the related work on Fuzzy C-Means (FCM), Hadoop, Cloudera, etc.; Section 3 discusses the English data set; Section 4 overviews the methodology of our proposed model; Section 5 describes the experiment and Section 6 provides the conclusion.

2 Related work

In this section, we overview several studies related to Fuzzy C-Means (FCM), the Vector Space Model, Hadoop, Cloudera, etc.

There are many studies which are related to the Vector Space Model [2–4]. First of all, the authors of [2] transfer all English sentences into many factors which are used in VSM algorithm. In this research, the authors examine the Vector Space Model, an information retrieval technique and its variations. The rapid growth of the World Wide Web and the abundance of documents and different forms of information available on it, has resulted in the need for better information retrieval techniques. The Vector Space Model is an algebraic model used for information retrieval. It represents a natural language document in a formal manner by the use of vectors in a multi-dimensional space, and allows decisions to be made as to which documents are similar to each other and to the queries fired. This work also explains the existing variations of the VSM and proposes a new variation that should be considered [3]. In the text classification task, one of the main problems is to choose which features give the best results. Various features can be used such as words, n-grams, syntactic n-grams of various types (POS tags, dependency relations, mixed, etc.), or a combination of these features. Also, algorithms to reduce the dimensionality of these sets of features can be applied, such as Latent Dirichlet Allocation (LDA). In this research, the authors consider the multi-label text classification task and apply various feature sets. The authors consider a subset of multi-labeled files of the Reuters-21578 corpus. The authors use traditional TF-IDF values of the features and tried both considering and ignoring the stop words. The authors also tried several combinations of features, like bi-grams and uni-grams. The authors also experimented by adding LDA results into Vector Space Models as new features. These latter experiments obtained the best results [4]. KNN and SVM are two machine learning approaches to text categorization (TC) based on the Vector Space Model. In this model, borrowed from information retrieval, documents are represented as a vector where each component is associated with a particular word from the vocabulary. Traditionally, each component value is assigned using the information retrieval TFIDF measure. While this weighting method seems very appropriate for IR, it is not clear that it is the best choice for TC problems. Actually, this weighting method does not leverage the information implicitly contained in the categorization task to represent documents. In this research, the authors introduce a new weighting method based on the statistical estimation of the importance of a word for a specific categorization problem. This method also has the benefit of making feature selection implicit, since useless features of the categorization problem considered are assigned a very small weight. Extensive experiments reported in the research show that this new weighting method significantly improves classification accuracy as measured on many categorization tasks.

Many studies such as [5-7] are related to the implementation of algorithms and applications in the parallel network environment. Hadoop is an Apache-based framework which is used to handle large data sets on clusters consisting of multiple computers, using the Map and Reduce programming model. The two main projects of Hadoop are the Hadoop Distributed File System (HDFS) and Hadoop M/R (Hadoop Map/Reduce). Hadoop M/R allows engineers to program for writing applications for the parallel processing of large datasets on clusters consisting of multiple computers. An M/R task has two main components: (1) Map and (2) Reduce. This framework splits the input data into chunks which multiple Map tasks can handle as a separate data partition in parallel. The outputs of the map tasks are gathered and processed by the Reduce task which is ordered. The inputs and outputs of each M/R are stored in HDFS because the Map tasks and the Reduce tasks are performed on the pair (key, value), and the formatted input and output formats will be the pair (key, value) [7]. Cloudera, the global provider of the fastest, easiest, and most secure data management and analytics platform built on ApacheTM Hadoop® and the latest open source technologies, announced in November 2015 that it will submit proposals for Impala and Kudu to join the Apache Software Foundation (ASF). By donating its leading analytic database and columnar storage projects to the ASF, Cloudera aims to accelerate the growth and diversity of their respective developer communities. Cloudera delivers the modern data management and analytics platform built on Apache Hadoop and the latest open source technologies. The world's leading organizations trust Cloudera to help solve their most challenging business problems with Cloudera Enterprise, the fastest, easiest and most secure data platform available currently. Cloudera's customers are able to efficiently capture, store, process and analyze vast amounts of data, empowering them to use advanced analytics to drive business decisions quickly, flexibly and at a lower cost than has been possible before. To ensure Cloudera's customers are successful, it offers comprehensive support, training and professional services.

There are many studies, such as [8–24] which are related to the FCM algorithm.

Many studies are related to FCM in parallel systems (or FCM in distributed systems) such as the work in [25–27].



Many studies, such as [28–42] are related to parallel systems or distributed systems.

Research using FCM for sentiment classification can be found in [43-50].

The latest research on sentiment classification can be found in [51-54, 56].

3 Data set

The English training data set includes 60,000 English sentences in the movie field, of which 30,000 are positive English sentences.

Fig. 2 Our english testing data set

All English sentences in our English training data set have been automatically extracted from Facebook and websites in social networks, after which we labeled them as either positive or negative. Figure 1 is the English training data set of this model.

We used a publicly available large data set of movie reviews from the Internet Movie Database [1]. This English data set comprises a testing data setwhich we refer to as the first testing data set and also a training data setwhich we refer to as the second testing data set. Both our first testing data set and our second testing data set contain 25,000 English documents, each with 12,500 positive English movie reviews and 12,500 negative English movie



reviews. Figure 2 is the English testing data set of this model.

4 Methodology

The methodology section comprises two parts: the semantic classification of the 25,000 English documents in the testing t1 and the 25,000 English documents of the testing t2 on the sequential environment is presented in the first part and the sentiment classification of the 25,000 English reviews of the testing t1 and the 25,000 English reviews of the testing t2 in the parallel network environment is presented in the second part.

In the English training data set, there are two clusters: the first, called the positive cluster, contains 30,000 positive English sentences and the second, called the negative cluster, contains 30,000 negative English sentences. All English sentences in both the first cluster and the second cluster have undergone word segmentation and stop-word removal after which they are transferred into vectors (vector representation). The 30,000 positive English sentences in the positive cluster are transferred into the 30,000 positive vectors, called the positive vector group (or the positive vector cluster). The 30,000 negative English sentences in the negative cluster are transferred into 30,000 negative vectors, called the negative vector group (or the negative vector cluster). Therefore, the English training data set includes the positive vector group (or the positive vector cluster) and the negative vector group (or the negative vector cluster) [2–4]. The VSM is an algebraic model used for information retrieval. It represents a natural language document in a formal manner by the use of vectors in a multidimensional space. The VSM is a way of representing documents through the words that they contain. Vector space modeling places terms, documents, and queries in a term-document space so it is possible to compute the similarities between queries and the terms or documents, and allow the results of the computation to be ranked according to the similarity measure between them. The VSM allows decisions to be made about which documents are similar to each other and to queries.

We transferred all the English sentences in the training data set into vectors similar to VSM [2–4].

4.1 Fuzzy C-means algorithm in the sequential environment

Figure 3 illustrates how sentiment classification is undertaken in the sequential environment.



With each English document in the English testing data set, we assume that each English document has n English sentences and we transfer the n English sentences into n vectors similar to VSM [2–4]. Thus, the document has n vectors. For each vector of the n vectors, we use FCM to cluster the vector into the positive vector group or the negative vector group in the sequential environment. According to [8–17], we implement the FCM algorithm which is enhanced to be able to classify the sentiment of the English sentences.

The total all the fuzzy partitions which have c clusters of N objects in D is calculated as follows:

$$E_{fc} = \left\{ U \in R_{cN} | \underset{1 \le i \le c \land 1 \le k \le N}{\forall} u_{ik} \in [0, 1], \sum_{i=1}^{c} u_{ik} \right.$$
$$= 1, 0 < \sum_{k=1}^{N} u_{ik} < N \right\}$$

Minimize the objective function:

$$J_m(U, V) = \sum_{i=1}^{c} \sum_{k=1}^{N} (u_{ik})^m d_{ik}^2$$
$$d_{ik}^2 = |x_k - v_i|_A$$

V = [v1, v2, ..., vc] is a matrix which represents the center object values of the cluster. A matrix is a positive finite. m is the exponent weight in $[1, \infty)$.

The objective function reaches a minimum value if and only if:

$$\forall I_{k} = \{i | 1 \le i \le c; d_{ik} = 0\}$$

$$\forall u_{ik} = \begin{cases} (d_{ik})^{\frac{2}{1-m}} \left[\sum_{j=1}^{c} (d_{ik})^{\frac{2}{1-m}} \right]^{-1} \\ 0, i \notin \\ \sum_{i \in I_{k}} u_{ik} = 1, i \in I_{k} \end{cases}$$

$$(1)$$

$$\bigvee_{1 \le i \le c} v_i = \frac{\sum_{k=1}^{N} (u_{ik})^m x_k}{\sum_{k=1}^{N} (u_{ik})^m}$$
(2)

The FCM algorithm comprises the following steps:

Algorithm 1 Fuzzy c-means algorithm

Input: One vector of the document of the testing data set; the positive vector group of the training data set and the negative vector group of the training data set.

Output: The result of clustering the vector into the positive vector group or the negative vector group.

Begin

Step 1: Enter values for the two parameters: c (1 <c <N), m and initializing the sample matrix

with $||U||_F^2 = \sum_i \sum_k U_{ik}^2$

With the clustering results of the n vectors of the documents in the testing data set, the document has a positive sentiment if the number of vectors in the n vectors is greater than the number of vectors in the n vectors. The document has a negative sentiment if the number of vectors in the n vectors is less than the number of vectors in the n vectors. The document has a neutral sentiment if the number of vectors in the n vectors is equal to the number of vectors in the n vectors.

4.2 Fuzzy C-means (FCM) in the parallel network environment

Figure 4 illustrates how semantic classification is undertaken in a parallel network environment.

We transfer the 60,000 English sentences in the training data set into the 60,000 vectors using Hadoop Map (M)/Reduce (R) in the Cloudera parallel network environment to shorten the execution time of this task. Figure 5 overviews the process of transferring each English sentence into one vector in the Cloudera networkenvironment.

Transferring each English sentence into one vector in the Cloudera network environment involves two phases: Map (M) phases and Reduce (R) phases. The input of the Map phase is one English sentence and the output of the Map phase are the many components of a vector which correspond to the sentence. In the Map phase of Cloudera, we transfer the sentence into one vector similar to VSM [2–4]. The input of the Reduce phase is the output of the Map phase, which is many components of a vector. The output of the Reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the reduce phase is a vector which corresponds to the vector similar to VSM [2–



Input of Hadoop Reduce in Cloudera: Output of Hadoop Map in Cloudera

sentence. In the Reduce phase of Cloudera, these components of the vector are built into one vector.

Each English document in the testing data set contains n English sentences. We transfer each English sentence in the n English sentences into one vector similar to the process shown in Fig. 5. Hence, the document also has n vectors.

FCM in the Cloudera parallel network environment comprises two phases: the first phase is the Hadoop Map (M) phase in Cloudera and the second phase is the Hadoop Reduce (R) phase in Cloudera. In the Map phase, the input is the n vectors of one English document (which have been classified) into either the positive vector group or the negative vector group; and the output is the clustering results of the n vectors of the document into either the positive vector group or the negative vector group. In the Reduce phase, the input is the output of the Map phase and this input is the clustering results of the n vectors of the document into either the positive vector group or the negative vector group; and the output is the sentiment classification result of the document as either having positive polarity, negative polarity, or neutral polarity. In the Reduce phase, the English document is classified as having a positive sentiment if the number of vectors of the n vectors in the positive vector group is greater than the number of vectors of the n vectors in the negative vector group; the English document is classified as having a negative sentiment if the number of vectors of the n vectors in the positive vector group is less than the number of vectors of the n vectors in the negative vector group; and the English document is classified as having a neutral sentiment

Fig. 7 Overview of Hadoop reduce (R) in Cloudera

if the number of vectors of the n vectors in the positive vector group is equal to the number of vectors of the n vectors in the negative vector group.

4.2.1 Hadoop Map (M)

Figure 6 illustrates the Hadoop Map phase.

Similar to [7-17], we propose FCM as follows:

Algorithm 2 Fuzzy c-means algorithm

Input: The n vectors of the document of the testing data set; the positive vector group of the training data set and the negative vector group of the training data set. Output: the result of clustering The n vectors into the positive vector group or the negative vector group. Begin Step 0: With each vector in the n vectors, repeat: Step 1: Enter values for the two parameters: c (1 < c < N), m and initializing the sample matrix Step 2: Repeat 2.1 i=i+1;2.2 Calculating fuzzy partition matrix Uj following formula(1)2.3 Updating centers $V(j) = [v_1(j), v_2(j), ..., v_c(j)]$ basing on (2) và Uj matrix; Step 3: Until ($||U_{(j+1)} - U_{(j)}||_F$ $\leq \xi$; Step 4: Performing results of the clusters. Step 5: End Step 0; End;



Table 1 The results of the25,000 english documents intesting data set t1		Testing dataset t1	Correct classification	Incorrect classification
	Negative	12500	7523	4977
	Positive	12500	7527	4973
	Summary	25000	15050	9950

4.2.2 Hadoop Reduce (R)

Figure 7 illustrates the Hadoop Reduce phase.

5 Experiment

We used measures such as accuracy (A) to calculate the accuracy of the results of sentiment classification.

The Java programming language was used to save the data sets in order to implement our proposed model to classify the 25,000 English documents in testing data set t1 and the 25,000 English documents of testing data set t2.

To implement the proposed model, we used the Java programming language to save the English training data set, the English testing data set and the results of the sentiment classification.

The sequential environment in this research comprises one node (one server). The Java language is used to program FCM. The configuration of the server in the sequential environment is Intel[®] Server Board S1200V3RPS, Intel[®] Pentium[®] Processor G3220 (3M Cache, 3.00 GHz), 2GB PC3-10600 ECC 1333 MHz LP Unbuffered DIMMs. The operating system of the server is Cloudera.

We implement FCM in the Cloudera parallel network environment - this Cloudera system comprises four nodes (four servers). The Java language is used to program the application of the FCM in Cloudera. The configuration of each server in the Cloudera system is Intel[®] Server Board S1200V3RPS, Intel[®] Pentium[®] Processor G3220 (3M Cache, 3.00 GHz), 2GB PC3-10600 ECC 1333 MHz LP Unbuffered DIMMs. The operating system of each of the four nodes is Cloudera. All nodes have the same configuration.

The results of the sentiment classification of the 25,000 English documents in testing data set t1 are presented in Table 1.

The results of the sentiment classification of the 25,000 English documents in testing data set t2 are presented in Table 2.

The accuracy of the sentiment classification of the 25,000 English documents in testing dataset t1 is shown in Table 3.

The accuracy of the sentiment classification of the 25,000 English documents in testing dataset t2 is shown in Table 4.

6 Conclusion

Although our proposed model was tested on an English data set, it can also be applied to many other languages. In this paper, our model was tested on the 25,000 English documents in the testing data set t1 and the 25,000 English documents in the testing data set t2 which are small data sets. However, our model can be applied to a big data set containing millions of English documents in a very short time.

In this work, we proposed a new model to classify the sentiments of English documents using the Fuzzy C-Means Algorithm (FCM) with Hadoop Map (M) /Reduce (R) in the Cloudera parallel network environment. The experiment results show that our proposed model achieves 60.2 % and 59.8 % accuracy of the English documents. Currently, there is a paucity of research which shows that clustering methods can be used to classify data. Our research shows that

Table 2The results of the25,000 english documents intesting data set t2

	Testing dataset t2	Correct classification	Incorrect classification
Negative	12500	7437	5063
Positive	12500	7363	5137
Summary	25000	14800	10200

Model	Class Accuracy		FCM Algorithm in the sequential environment	FCM Algorithm in the Cloudera distributed system		
Our proposed model	Negative	60.2 %	Average time of the classifi- cation: 150,590 seconds/25,000 English documents	Average time of the classifi- cation: 37,659 seconds/25000 English documents		
	Positive					

Table 3 The accuracy of our proposed model for the sentiment classification of the 25,000 english documents in testing data set t1

clustering methods are able to classify data and in particular, they are useful for sentiment classification for text.

As shown in Table 3, the average time taken for the sentiment classification of the 25,000 English documents in testing data set t1 using the FCM algorithm in the sequential environment is 150,590 seconds, which is greater than the average time taken for the sentiment classification of the 25,000 English documents using FCM in the Cloudera parallel network environment, which is 37,659 seconds.

As shown in Table 4, the average time taken for the sentiment classification of the 25,000 English documents in testing data set t2 using the FCM algorithm in the sequential environment is 151590 seconds, which is greater than the average time taken for the sentiment classification of the 25,000 English documents in testing data set t2 using FCM in the Cloudera parallel network environment, which is 37875 seconds.

The execution time of the FCM in Cloudera is dependent on the performance of the Cloudera parallel system and is also dependent on the performance of each server on the Cloudera system.

The principles underpinning our proposed model for classifying the sentiment (positive, negative, neutral) of the English documents in the English testing data set in the sequential environment, based on the numerous English sentences in the English training data set are similar to the principles underpinning our proposed model for classifying the sentiment (positive, negative, neutral) of the English documents in English testing data set in the distributed environment, based on the numerous English sentences in English training data set.

The FCM of our proposed model in the sequential environment is different from the FCM of our proposed model in the parallel environment. We built many algorithms related to the FCM to implement our model in the distributed system.

The execution time of our model in the parallel environment is less than the execution time of our model in the sequential environment. The execution of our model in the distributed system is shorter if the performance in the distributed system is longer.

In addition, the execution time of any model is also dependent on the algorithms. For example, using the same algorithms, different systems perform differently and have different execution times. Using the same system with the same performance, different algorithms may have different execution times.

Our survey has many advantages and disadvantages. The advantages are: it processes big data involving millions of English documents; the execution time of our model to conduct sentiment on big data is short, etc. However, the disadvantages are: it takes a long time to implement and it is costly to build the algorithms of the model in the distributed system.

Model	Class	Accuracy	FCM Algorithm in the sequential environment	FCM Algorithm in the Cloudera distributed system
Our proposed model	Negative	59.8 %	Average time of the classifi- cation: 151,590 seconds/25,000 English documents	Average time of the classifi- cation: 37,875 seconds/25,000 English documents
	Positive			

Table 4 The accuracy of our proposed model for the sentiment classification of the 25,000 english documents in testing data set t2

Table 5 Comparison of our model's results with the work in [2–4]

Studies	FCM	СТ	SC	PNS	SD	DT	L	VSM	
[2]	No	No	No	No	Yes	No	EL	Yes	
Model/method in [2]	Examining	the Vector Spa	ace Model, an i	nformation retri	eval technique a	and its variations	S		
Summary of the work in [2]	In this work The rapid g able on it h algebraic m the use of v ilar to each variation th	c, the authors rowth of Worl has resulted in odel used for i ectors in a mu other and to the at should be co	examine the Ve d Wide Web an a the need for l information retr lti-dimensional he queries fired onsidered	ctor Space Mod d the abundance better information rieval. It represent space, and allow . It also explains	lel, an informati e of documents on retrieval tech nts a natural lan ws decisions to s the existing va	on retrieval tech and different for hniques. The Va guage document be made as to w riations of the V	nnique and its rms of inform ector Space M t in a formal n thich documen /SM and prop	variations. ation avail- lodel is an nanner with nts are sim- poses a new	
[3]	No	No	Yes	No	Yes	No	EL	Yes	
Model/method in [3]	+Latent D +Several co	+Latent Dirichlet Allocation (LDA). +Multi-label text classification task and applies various feature sets. +Several combinations of features, like bi-grams and uni-grams.							
Summary of the work in [3]	+ The auth Vector Space	+ The authors proposed the use of a VSM complement based on the LDA, adding the LDA results as features to Vector Space Models							
[4]	No	No	Yes	No	Yes	Yes	Yes	Yes	
Model/method in [4]	KNN and S	VM are two m	achine learning	approaches to to	ext categorizatio	on (TC) based or	n the Vector Sp	pace Model	
Summary of the work in [4]	+In this model, borrowed from information retrieval, documents are represented as vectors where each compo- nent is associated with a particular word from the vocabulary. Traditionally, each component value is assigned a weight using the information retrieval TFIDF measure. While this weighting method seems very appropriate for IR, it is not clear that it is the best choice for TC problems. Actually, this weighting method does not leverage the information implicitly contained in the categorization task to represent documents. In this work, the authors introduce a new weighting method based on the statistical estimation of the importance of a word for a specific categorization problem. This method also has the benefit of making feature selection implicit, since the useless features of the categorization problem are assigned a very small weight. Extensive experiments reported in the work show that this new weighting method significantly improves classification accuracy as measured on many categorization tasks. + In this work, the authors presented a new method to weight features in the vector-space model for text categorization by leveraging the categorization task. The most commonly used method is TFIDF,								
Our work	Yes	Yes	Yes	Yes	Yes	Yes	EL	Yes	
Model/Method in our work	Fuzzy C-M	eans algorithn	n for English se	ntiment classific	cation in the Clo	oudera distribute	ed system		
Summary of our work	Firstly, we polarity, ne algorithm (1 polarity in t	use the Fuzzy gative polarity FCM) to class he Cloudera d	C-Means algo or neutral pol- ify the English istributed envir	orithm (FCM) to arity in the seque documents as has onment with the	o classify Engl uential environm aving either pos e purpose of sho	ish documents a nentThen, we itive polarity, ne ortening the exec	as having eith use the Fuzz gative polarity cution time	er positive y C-Means y or neutral	

To understand the scientific value of this research, we compare our model's results with the results of models used in other studies.

Table 5 compares our model's results with the studies in [2-4] as follows:

cluster technique: CT.

sentiment classification: SC (opinion mining, or semantic classification, or emotion classification). parallel network system: PNS (distributed system). special domain: SD. dependence on the training data set: DT. language: L

Vector Space Model: VSM

no mention: NM English language: EL. Fuzzy C-Means: FCM.

Table 6 Compares our model's results with the work related to the Fuzzy C-Means (FCM) algorithm in [8–24].

Table 7 compares our model's results with studies related to Fuzzy C-Means in the parallel system (or FCM in the distributed system) in [25-27].

Table 8 compares our model's results with studies related to FCM for sentiment classification in [43-50].

Table 9 compares our model's results with the latest research on sentiment classification (or sentiment analysis or opinion mining) in [51-56].

Table 6	Comparison	of our model's	results with the	e work related to	the Fuzzy	C-Means (F	FCM) algorithm	in [8–24]
---------	------------	----------------	------------------	-------------------	-----------	------------	----------------	-----------

Work	FCM	СТ	SC	PNS	SD	DT	L	VSM			
[8]	Yes	Yes	No	No	Yes	Yes	NM	NM			
Model/Method in [8]	Ambiguity	-driven fuzzy (C-means cluster	ing							
Summary of the work in [8]	As a well-k one cluster subject to f some cases security an outcomes. ² evidence to records in F the detected accuracy b preventing proposed m improved s	nown clusterin , providing m alse detections . The false det d medical diag They mainly e o make a good FCM by introd d ambiguous r y lowering the performance 1 nethod on seve ensitivity of th	ng algorithm, F ore flexibility t is caused by nois ections are very gnosis, where we merge from ma decision. In th ucing a certaint ecords to anothe error rate. Mo oss which is co ral data sets fro e algorithm	uzzy C-Means of han non-fuzzy sy records, wea y important in s veak decisions la king decisions is survey, the a y factor to decro- er discrimination ost of the record mmon in simila m different dom	(FCM) allows e clustering meth k feature select ome decision-n based on such f about a subset uthors propose ease invalid dete on method for a ds are still proc ar hybrid methon nains show a sig	ach input sampl ods. However, ion and low cer- naking applicati alse detections of records that a method for d ections. This app deeper investig essed quickly a ds. Experiment gnificant decrea	le to belong to the accuracy tainty of the a on domains li may lead to c do not provid etecting such proach enables ation, thus inc und with a low al results of a se in error rate	o more than of FCM is lgorithm in ke network atastrophic e sufficient ambiguous s us to send creasing the v error rate pplying the e as well as			
[9]	Yes	Yes	No	No	Yes	Yes	NM	NM			
Model/Method in [9]	Generalized	d fuzzy c-meai	ns clustering str	ategies							
Summary of the work in [9]	Fuzzy c-me robustness case where (though con facilitate th in a numeri	uzzy c-means (FCM) is a useful clustering technique. Modifications of FCM using L_1 norm distances increase obustness to outliers. Object and relational data versions of FCM clustering are defined for the more general ase where the Lp norm or semi-norm (0 <p<1) as="" authors="" dissimilarity.="" give="" is="" measure="" of="" simple<br="" the="" used="">hough computationally intensive) alternating optimization schemes for all object data cases of p>0 in order to acilitate the empirical examination of the object data models. Both object and relational approaches are included in a numerical study</p<1)>									
[10]	Yes	Yes	No	No	Yes	Yes	NM	NM			
Model/Method in [10]	Fuzzy Koh	onen clustering	g networks								
Summary of the work in [10]	Kohonen n set of heuri guaranteed of data). A into the lea related to F proved that Kohonen ty the compet method; an	Kohonen networks are well known for cluster analysis (unsupervised learning). This class of algorithms is a set of heuristic procedures that suffers from several major problems (e.g. neither termination or convergence is guaranteed, no model is optimized by the learning strategy, and the output is often dependent on the sequence of data). A fuzzy Kohonen clustering network is proposed which integrates the Fuzzy c-Means (FCM) model into the learning rate and updating the strategies of the Kohonen network. This yields an optimization problem related to FCM, and the numerical results show improved convergence as well as reduced labeling errors. It is proved that the proposed scheme is equivalent to the c-Means algorithms. The new method can be viewed as a Kohonen type of FCM, but is "self-organizing" since the "size" of the update neighborhood and learning rate in the competitive layer are automatically adjusted during learning. Anderson's IRIS data is used to illustrate this									
[11]	Yes	Yes	No	No	Yes	Yes	NM	NM			
Model/Method in [11]	Fuzzy c-me	eans clustering	of incomplete	data							
Summary of the work in [11]	The problet observation data sets ca x(k) might values are r it is not dire data sets ar properties of incomplete	The problem of clustering or incomplete data The problem of clustering a real s-dimensional data set $X=\{x(1),,x(n)\}$ subset $R(s)$ is considered. Usually, each observation (or datum) consists of numerical values for all s features (such as height, length, etc.), but sometimes data sets can contain vectors that are missing one or more of the feature values. For example, a particular datum x(k) might be incomplete, having the form $x(k)=(254.3, ?, 333.2, 47.45, ?)(T)$, where the second and fifth feature values are missing. The fuzzy c-means (FCM) algorithm is a useful tool for clustering real s-dimensional data, but it is not directly applicable to the case of incomplete data. Four strategies for doing FCM clustering of incomplete data sets are given, three of which involve the modified versions of the FCM algorithm. Numerical convergence properties of the new algorithms are discussed, and all approaches are tested using real and artificially generated									
[12]	Yes	Yes	No	No	Yes	Yes	NM	NM			
Model/Method in [12]	The color i	mage segment	ation algorithm	based on the th	resholding and	the fuzzy c-mea	ins techniques				
Summary of the work in [12]	In this rese (FCM) tech component FCM. The segmentation using the F existing alg and the resise most accur amount of c	earch, a segme iniques is pres- s. The method coarse segmen on assigns the CM. Attempts gorithms—Ohl ults are discus- ate segmented computational	ntation algorith ented. The scale ology uses a co- ntation attempts pixels, which also have been ander's, Rosenf sed in this pape image on the o effort	am for color im- e-space filter is to parse-fine concess is to segment co- remain unclassi- made to compa feld's, and Bezd r. The simulatic color coordinate	ages based on used as a tool for ept to reduce the arsely using the ified after the correct the performa- ek's. Intensive of n results indicate proposed by 0	the thresholding or analyzing the e computationa e thresholding to oarse segmenta nce of the propo- computer simula- te that the prop- Dhta et al., whi	g and the fuzz histograms of l burden requi technique, wh ttion, to the cl osed algorithm ation has been osed algorithm le requiring a	zy c-means three color ired for the ile the fine lossest class a with other performed n yields the reasonable			

Table 6(continued)

Work	FCM	СТ	SC	PNS	SD	DT	L	VSM		
[13]	Yes	Yes	NM	NM	NM	NM	NM	NM		
Model/Method in [13] Summary of the work in [13]	A FORTRA This paper is applicabl and prototy or suggestin least-square Mahalonob numbers of	AN-IV coding of develop a FOR le to a wide vau ypes for any se ng substructure es objective fur bis), an adjustal clusters, and c	of the fuzzy c-1 TRAN-IV cod iety of Geo-sta t of numerical s in unexplored ction. Features ble weighting fr outputs that inc	neans (FCM) cl ing of the fuzzy atistical data and data. These par d data. The cluss of this program actor that essent lude several me	lustering program c-means (FCM alysis problems, titions are usefu- tering criterion u- n include a choic tially controls se asures of cluster	m.) clustering pro- This program g 11 for corroborat used to aggregat e of three norms ensitivity to nois r validity	gram. The FC generates fuzz, ting known su e subsets is a g (Euclidean, E se, acceptance	M program y partitions ibstructures generalized Diagonal, or of variable		
[14]	Yes	Yes	NM	NM	NM	NM	NM	NM		
Model/Method of [14]	Many func (FCM) clus	tions have bee stering algorith	n proposed for m	validation of p	partitions of obj	ect data produce	ed by the fuzz	zy c-means		
Summary of [14]	Many func (FCM) clus nent m of t the partitio and the Ful Fukuyama- the indexes clusters, (2 in the inter- FCM	lany functions have been proposed for validation of partitions of object data produced by the fuzzy c-means ³ CM) clustering algorithm. The authors examine the role a subtle but important parameter-the weighting expo- ent m of the FCM model-plays in determining the validity of FCM partitions. The functionals considered are a partition coefficient and entropy indexes of Bezdek, the Xie-Beni (1991), and extended Xie-Beni indexes, nd the Fukuyama-Sugeno index (1989). Limit analysis indicates, and numerical experiments confirm, that the ukuyama-Sugeno index is sensitive to both high and low values of m and may be unreliable because of this. Of the indexes tested, the Xie-Beni index provided the best response over a wide range of choices for the number of usters, (2–10), and for m from 1.01-7. The authors' calculations suggest that the best choice form is probably the interval [1.5, 2.5], whose mean and midpoint, m=2, have often been the preferred choice for many users of								
[15]	Yes	Yes	NM	NM	NM	NM	NM	NM		
Model/Method in [15]	A new mod	lel called possi	bilistic-fuzzy c	-means (PFCM) model					
Summary of the work in [15]	The authors ship and ty over all dat for large da produces m for each clu avoids the comes the derive the basis for a s Several nur	The authors proposed the fuzzy-possibilistic c-means (FPCM) model and algorithm that generates both member- ship and typicality values when clustering unlabeled data. FPCM constrains the typicality values so that the sum over all data points of typicalities to a cluster is one. The row sum constraint produces unrealistic typicality values for large data sets. The authors proposed a new model called possibilistic-fuzzy c-means (PFCM) model. PFCM produces memberships and possibilities simultaneously, along with the usual point prototypes or cluster centers for each cluster. PFCM is a hybridization of possibilistic c-means (PCM) and fuzzy c-means (FCM) that often avoids the various problems of PCM, FCM and FPCM. PFCM solves the noise sensitivity defect of FCM, over- comes the coincident clusters problem of PCM and eliminates the row sum constraints of FPCM. The authors derive the first-order necessary conditions for extrema of the PFCM objective function, and use them as the basis for a standard alternating optimization approach to finding local minima of the PFCM objective functional.								
[16]	Yes	Yes	NM	NM	NM	NM	NM	NM		
Model/Method in [16]	A novel alg inhomogen	orithm for fuzz eities using fuz	zy segmentatio zzy logic	n of magnetic re	esonance imagir	ng (MRI) data ai	nd estimation	of intensity		
Summary of the work in [16]	The authors the estimat to imperfect is a slowly classification c-means (F to be influe biases the s corrupted b	s present a nove ion of intensity tions in the race varying shadi on. The author (CM) algorithm enced by the la solution toward by salt and pepp	el algorithm for v inhomogeneit io-frequency c ng artifact ove s' algorithm is n to compensat bels in its imm piecewise-hom ver noise	r the fuzzy segn ties using fuzzy oils or to proble or the image that formulated by e for such in ho ediate neighbor nogeneous label	nentation of mag logic. MRI into ems associated v at can produce of modifying the omogeneities and hood. The neigh lings. Such a reg	gnetic resonance ensity inhomoge with the acquisit errors with com- objective functi d to allow the la hborhood effect gularization is us	imaging (MR eneities can be ion sequences ventional inter on of the stan abeling of a pi acts as a regu seful in segme	(1) data and e attributed . The result nsity-based idard fuzzy ixel (voxel) ilarizer and enting scans		
[17]	Yes	Yes	NM	NM	NM	NM	NM	NM		
Model/Method in [17]	An approxi the FCM ec	mate fuzzy c-n quation with in	neans (AFCM) teger-valued or	implementation real-valued est	n based upon rej imates	placing the nece	ssary "exact"	variates in		
Summary of the work in [17]	This research algorithms, tation based estimates. T tances and iteration is while appar numerically tions orient feature space	ch reports the r In particular, i d upon replacin This approxima for exponentia reduced to app rently preservin y on a nine-bar ted readers. Ou ce is comprised	esults of a num- he authors pro g the necessary ation enables A ation. The net roximately one ng the overall q d digital image r results sugge l of tuples havi	erical compariso pose and exemp v "exact" variate AFCM to exploi effect of the p e sixth of the tin uality of termin e, and a pseudo- st that AFCM m ng a finite numb	on of two version olify an approximation is in the FCM equivalent that a lookup table roposed implement roposed implement roposed implement al clusters produ- code subroutine may be used to a ber of integer-value	ns of the fuzzy c mate fuzzy c-me uation with inte e approach for c nentation is tha a literal implement aced. The two im- e is given for the accelerate FCM alued coordinate	-means (FCM eans (AFCM) ger-valued or omputing Euc t CPU time d entation of the nplementation e convenience processing wites) clustering implemen- real-valued clidean dis- during each e algorithm, is are tested of applica- henever the		

Table 6(continued)

(, , , , , , , , , , , , , , , , , , ,											
Work	FCM	СТ	SC	PNS	SD	DT	L	VSM			
[18]	Yes	Yes	NM	NM	NM	NM	NM	NM			
Model/Method in [18]	This countere than the geom	xample estanetric centroi	blishes the exis	tence of saddle artition space	points of the F	CM objective fu	nction at loca	ations other			
Summary of the work in [18]	A counterexain rithms (see J.C. This countere than the geom (1987) are sum least along a sum rucker's count propagate the	counterexample to the original incorrect convergence theorem for the fuzzy c-means (FCM) clustering algo- thms (see J.C. Bezdak, IEEE Trans. Pattern Anal. and Math. Intell., vol.PAMI-2, no.1, pp.1-8, 1980) is provided. 'his counterexample establishes the existence of saddle points of the FCM objective function at locations other nan the geometric centroid of fuzzy c-partition space. Counterexamples previously discussed by W.T. Tucker 1987) are summarized. The correct theorem is stated without proof: every FCM iterate sequence converges, at east along a subsequence, to either a local minimum or saddle point of the FCM objective function. Although 'ucker's counterexamples and the corrected theory appear elsewhere, they are restated as a caution not to further ropagate the original incorrect convergence statement									
[19]	Yes	Yes	NM	NM	NM	NM	NM	NM			
Model/Method in [19]	The authors su	ubstantially	improve RFCM	by generalizin	g it to the case o	f arbitrary (symi	metric) dissin	uilarity data			
Summary of the work in [19]	The relational dissimilarity of match the giv to most relation arbitrary (sym fication of the data. While the convert similar are positive, r approach to p	he relational fuzzy c-means (RFCM) algorithm can be used to cluster a set of n objects described by pair-wise ssimilarity values if (and only if) there exist n points in $Rn - 1$ whose squared Euclidean distances precisely tach the given dissimilarity data. This strong restriction on the dissimilarity data renders RFCM inapplicable o most relational clustering problems. This work substantially improves RFCM by generalizing it to the case of bitrary (symmetric) dissimilarity data. The generalization is obtained using a computationally efficient modi- cation of the existing algorithm that is equivalent to applying a "spreading" transformation to the dissimilarity ata. While the method given applies specifically to dissimilarity data, a simple transformation can be used to onvert similarity relations into dissimilarity data, so the method is applicable to any numerical relational data that re positive, reflexive (or antireflexive) and symmetric. Numerical examples illustrate and compare the present purpore to problems that can be studied with alternatives such as the linkage algorithms.									
[20]	Yes	Yes	NM	NM	NM	NM	NM	NM			
Model/Method in [20]	A novel algor	ithm for the	fuzzy segmenta	ation of magnet	ic resonance im	aging (MRI) dat	a				
Summary of the work in [20]	a novel algori by modifying distance metri FCM is replac kernelized fuz is added to the allow the labe and has a coef that the propo algorithms	thm for fuzz the objectiv ic and a spat ced by a kerr zzy C-means e objective fi ling of a pix fficient rang sed algorithi	e function for a generation e function in th ial penalty on t nel-induced dis (KFCM) algor unction in KFC el to be influen ing from zero to ms have better p	in many incure of magnetic re e conventional he membership tance, and thus ithm, which is s 'M to compensa ced by its neigh o one. Experim- performance wh	the imaging applied esonance imaging fuzzy C-means (o functions. First the correspondi hown to be more the for the intens abors in the imagental results on hen noise and oth	(FCM) algorithm (FCM) algorithm ly, the original I ng algorithm is probust than FCI ity inhomogenei ge. The penalty to both synthetic an her artifacts are p	Fine algorithm n using a kern Euclidean dist derived and c M. Then a spa ities of MR in term acts as a nd real MR in present than t	is realized nel-induced tance in the alled as the tial penalty nage and to regularizer nages show he standard			
[21]	Yes	Yes	NM	NM	NM	NM	NM	NM			
Model/Method in [21]	New relationa	l versions of	f the hard and f	uzzy c-means a	lgorithms						
Summary of the work in [21]	The hard and into (hard or f feature vector described in to pairs of object when the relat the algorithms	The hard and fuzzy c-means algorithms are widely used, effective tools for the problem of clustering n objects into (hard or fuzzy) groups of similar individuals when the data is available as object data, consisting of a set of n feature vectors in R. However, object data algorithms are not directly applicable when the n objects are implicitly described in terms of relational data, which consists of a set of n measurements of relations between each of the pairs of objects. New relational versions of the hard and fuzzy c-means algorithms are presented here for the case when the relational data can reasonably be viewed as some measure of distance. Some convergence properties of									
[22]	Yes	Yes	NM	NM	NM	NM	NM	NM			
Model/Method in [22]	A fuzzy c-m function for c	eans (FCM) lustering) algorithm the	at incorporates	spatial inform	ation into the	membership				
Summary of the work in [22]	A conventiona present a fuzz clustering. Th under conside than those of a to noise than a both single an	al FCM algo y c-means (l e spatial fur ration. The a other method other technic ad multiple-f	rithm does not f FCM) algorithm action is the sun advantages of th ls, (2) it reduces ques. This techn eature data with	fully utilize the In that incorpora Inmation of the le new method a s the spurious b hique is a powe h spatial inform	spatial informati tes spatial infor membership fur re the following lobs, (3) it remover ful method for ation	on in the image. mation into the mation in the nei : (1) it yields reg ves noisy spots, noisy image seg	In this work, membership f ghborhood of gions more ho and (4) it is le mentation and	the authors function for f each pixel mogeneous ss sensitive d works for			

Table 6(continued)

Work	FCM	СТ	SC	PNS	SD	DT	L	VSM		
[23]	Yes	Yes	No	No	NM	Yes	NM	No		
Model/Method in [23]	A new weighted	l and constrain	ed possibilistic	C-means cluster	ring algorithm					
Summary of the work in [23]	In this work, a r fault detection a bilistic clusterin algorithm leadin tion in FCM cos without having simultaneously ferent weights t essential charac scale Tennessee the superiority of approaches	n this work, a new weighted and constrained possibilistic C-means clustering algorithm is proposed for process ault detection and diagnosis (FDI) in offline and online modes for both already known and novel faults. A possi- ilistic clustering based approach is utilized here to address some of the deficiencies of the fuzzy C-means (FCM) lgorithm leading to more consistent results in the context of the FDI tasks by relaxing the probabilistic condi- ion in FCM cost function. The proposed algorithm clusters the historical data set into C different dense regions vithout having precise knowledge about the number of the faults in the data set. The algorithm incorporates imultaneously possibilistic algorithm and local attribute weighting for time-series segmentation. This allows dif- erent weights to be allocated to different features responsible for the distinguished process faults which is an assential characteristic of proper FDI operations. A set of comparative studies have been carried out on the large- cale Tennessee Eastman industrial challenge problem and the DAMADICS actuator benchmark to demonstrate he superiority of the proposed algorithm in process FDI applications with respect to some available alternative approaches								
[24]	Yes	Yes	No	No	NM	Yes	NM	No		
Model/Method in [24]	Clustering Incom	mplete Data U	sing Kernel-Ba	sed Fuzzy C-me	ans Algorithm					
Summary of the work in [24]	There is a recent using the 'kerned fisher discrimin using kernel me feature space, w clustering proto using additional this work, a nove (FCM) is propor metric in the dat in the data space analysis shows property is utili KFCM has bette clustering	There is a recent trend in recent machine learning community to construct a nonlinear version of a linear algorithm using the 'kernel method', e.g. support vector machines (SVMs), kernel principal component analysis, kernel fisher discriminant analysis and the recent kernel clustering algorithms. In unsupervised clustering algorithms using kernel method, typically, a nonlinear mapping is used first to map the data into a potentially much higher feature space, where clustering is then performed. A drawback of these kernel clustering algorithms is that the clustering prototypes lie in high dimensional feature space and hence lack clear and intuitive descriptions unless using additional projection approximation from the feature to the data space as done in the existing literatures. In this work, a novel clustering algorithm using the 'kernel method' based on the classical fuzzy clustering algorithm (FCM) is proposed and called the kernel fuzzy c-means algorithm (KFCM). KFCM adopts a new kernel-induced metric in the data space to replace the original Euclidean norm metric in FCM and the clustered prototypes still lie in the data space so that the clustering results can be reformulated and interpreted in the original space. Authors' analysis shows that KFCM is robust to noise and outliers and also tolerates unequal sized clusters. Finally this property is utilized to cluster incomplete data. Experiments on two artificial and one real datasets show that								
Our work	Yes	Yes	Yes	Yes	Yes	Yes	English	Yes		
Model/Method in our work	Fuzzy C-Means	algorithm for	English sentim	ent classification	n in the Cloude	ra distributed s	ystem			
Summary of our work	Firstly, we use Fuzzy C-Means algorithm (FCM) to classify the English documents as having either positive polarity, negative polarity, or neutral polarity in the sequential environment. Then, we use the Fuzzy C-Means algorithm (FCM) to classify the English documents as having either positive polarity, negative polarity, or neutral polarity in the Cloudera distributed environment with the purpose of shortening the execution time									

Table 7	Comparison of our model's results with studies related to Fuzzy C-Means in the parallel system (or FCM in the distributed system) in
[25–27]	

Studies	FCM	CT	SC	PNS	SD	DT	L	VSM	
[25]	Yes	Yes	NM	NM	NM	NM	NM	NM	
Model/Method in [25]	The literal and neural network	he literal and approximate fuzzy c-means unsupervised clustering algorithms, and a supervised computational eural network							
Summary of the work in [25]	Magnetic reson resentations of clustering algo on normal volu- vised segment observed to sh complex segme lar MR relaxat being slightly p supervised vers	nance (MR) the original orithms, and unteers and ation technic ow better se entation prob ion behavior preferred over sus unsuperv	brain section i data with threa a supervised of selected patie gues provide gmentation we blem with turn , inconsistence er feedforward ised learning,	images are segm ee approaches: ti computational m nts with brain ti broadly similar then compared w nor/edema or cer y in rating amon d cascade correlat time complexity	ented and then he literal and ap eural network. umors surround results. Unsup with raw image ebrospinal fluid g experts was o tion results. Va y, and utility for	synthetically co pproximate fuzz The initial clini led by edema. S ervised fuzzy a data for volun boundary, whe observed, with f rious facets of b the diagnostic	lored to give cy c-means un cal results are Supervised an Igorithms we teer studies. I ere the tissues uzz-c-means both approach process, are c	visual rep- isupervised e presented id unsuper- ere visually For a more have simi- approaches ies, such as compared	
[26]	Yes	Yes	NM	NM	NM	NM	NM	NM	
Model/Method in [26]	Three clusterin the authors' pre-	Three clustering methods: (1) the conventional two-stage method, (2) the self-organizing feature maps, and (3) the authors' proposed two-stage method, via both simulated and real-world data							
Summary of the work in [26]	Cluster analysis ate analysis pro- have also been (1) the conven- stage method, self-organizing is slightly bett real-world data	Cluster analysis is a common tool for market segmentation. Conventional research usually employs the multivari- ate analysis procedures. In recent years, due to their high performance in engineering, artificial neural networks have also been applied in the area of management. Thus, this study aims to compare three clustering methods: (1) the conventional two-stage method, (2) the self-organizing feature maps and (3) the authors' proposed two- stage method, via both simulated and real-world data. The proposed two-stage method is a combination of the self-organizing feature maps and the K-means method. The simulation results indicate that the proposed scheme is slightly better than the conventional two-stage method with respect to the rate of misclassification, and the							
[27]	Yes	Yes	NM	Yes	NM	NM	NM	NM	
Model/Method in [27]	The parallel fu	zzy c-means	(PFCM) algo	rithm for cluster	ing large data s	ets			
Summary of the work in [27]	The parallel fu posed algorithm type with the l lel k-means (P. PFCM to clust almost ideal sp	The parallel fuzzy c-means (PFCM) algorithm for clustering large data sets is proposed in this survey. The pro- posed algorithm is designed to run on parallel computers of the Single Program Multiple Data (SPMD) model type with the Message Passing Interface (MPI). A comparison is made between PFCM and an existing paral- lel k-means (PKM) algorithm in terms of their parallelisation capability and scalability. In an implementation of PFCM to cluster a large data set from an insurance company, the proposed algorithm is demonstrated to have almost ideal speedups as well as an excellent scaleup with respect to the size of the data sets							
Our work	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Model/Method in our work	Fuzzy C-Mean	s algorithm	for English se	ntiment classific	ation in the Clo	oudera distribute	d system		
Summary of our work	Firstly, we use polarity, negati	the Fuzzy C ve polarity c	-Means algor r neutral pola	ithm (FCM) to c rity in the sequen	lassify the Eng	lish documents nt	as having eith	her positive	
	Then, we use tive polarity, n shortening the	the Fuzzy C egative pola execution tir	-Means algor rity or neutral ne	ithm (FCM) to l polarity in the	classify the En Cloudera distri	glish document buted environm	as as having e nent with the	either posi- purpose of	

Table 8 Comparison of our model's results with the FCM used for sentiment classification in [43–50]

Work	FCM	СТ	SC	PNS	SD	DT	L	VSM			
[43]	Yes	Yes	Yes	NM	NM	NM	NM	NM			
Model/Method in [43]	EmoSenticSp SenticNet by	EmoSenticSpace, a new framework for affective common-sense reasoning that extends WordNet-Affect and SenticNet by providing both emotion labels and polarity scores for a large set of natural language concepts									
Summary of the work in [43]	Emotions play a key role in natural language understanding and sensemaking. Pure machine learning usually fails to recognize and interpret emotions in text accurately. The need for knowledge bases that give access to semantics and sentics (the conceptual and affective information) associated with natural language is growing exponentially in the context of big social data analysis. To this end, this survey proposes EmoSenticSpace, a new framework for affective common-sense reasoning that extends WordNet-Affect and SenticNet by providing both emotion labels and polarity scores for a large set of natural language concepts. The framework is built by means of fuzzy c-means clustering and support-vector-machine classification, and takes into account a number of similarity measures, including point-wise mutual information and emotional affinity. EmoSenticSpace was tested on three emotion-related natural language processing tasks, namely sentiment analysis, emotion recognition, and personality detection. In all cases, the proposed framework outperforms the state-of-the-art. In particular, the direct evaluation of EmoSenticSpace against psychological features provided in the benchmark ISEAR dataset shows a 92.15 % agreement										
[44]	Yes	Yes	Yes	NM	NM	NM	NM	NM			
Model/Method in [44]	The task of se a much larger	mi-supervised of set	classificatio	n: extending cate	egory labels from	m a small datas	et of labeled e	examples to			
Summary of the work in [44]	The authors consider the task of semi-supervised classification: extending category labels from a small dataset of labeled examples to a much larger set. The authors show that, at least in their case study task, unsupervised fuzzy clustering of the unlabeled examples helps in obtaining the hard clusters. Namely, the authors used the membership values obtained with fuzzy clustering as additional features for hard clustering. The authors also used these membership values to reduce the confusion set for the hard clustering. As a case study, the authors use applied the proposed method to the task of constructing a large emotion lexicon by extending the emotion labels from the WordNet Affect lexicon using various features of words. Some of the features were extracted from the emotional statements of the freely available ISEAR dataset; other features were WordNet distance and the similarity measured via the polarity scores in the SenticNet resource. The proposed method classified words										
[45]	Yes	Yes	Yes	NM	NM	NM	NM	NM			
Model/Method in [45]	Techniques th	at have been us	ed for the ta	sk of opinion m	ining						
Summary of the work in [45]	As individuals impart their sentiments on the Web on products and services they have used, it has become impor- tant to formulate methods to automatically classify and judge them. The task of examining such data, collectively called client feedback data, is known as opinion mining. Opinion mining consists of several steps, and different techniques have been proposed for different steps. This survey basically explains such techniques that have been used for the implementation of task of opinion mining. On the basis of this analysis the authors provide an overall system design for the development of opinion mining approach										
[46]	Yes	Yes	Yes	NM	NM	NM	NM	NM			
Model/Method in [46]	Core concepts	s and techniques	s in the large	e subset of cluste	er analysis.						
Summary of the work in [46]	The fast retrieval of relevant information from databases has always been a significant issue. Many techniques have been developed for this purpose, of which data clustering is one of the major techniques. The process of creating vital information from the huge amount of data is learning, which can be classified as either supervised learning or unsupervised learning. Clustering is a kind of unsupervised data mining technique. It describes the general working behavior, the methodologies followed by these approaches and the parameters which affect the performance of these algorithms. In classifying web pages, the similarity between web pages is a very important feature. The main objective of this survey is to gather more core concepts and techniques in the large subset of cluster analysis.										
[47]	Yes	Yes	NM	NM	NM	NM	NM	NM			
Model/Method in [47]	A novel comb	vination of fuzzy	inference s	system and Dem	pster–Shafer Th	neory					
Summary of the work in [47]	Brain Magnet structure of the methods base this study, a m for the purpose proposed moor work is that the	tic Resonance I orain tissues as d on fuzzy appr ovel combinations be of segmentation deling, the consecution he rules are para	maging (M well as into oaches have on of fuzzy on where the equent part of aphrased as	RI) segmentation ensity non-unifor been developed inference system he pixel intensity of rules is a Dem evidences. The	n is a challengi rmity, partial v to overcome th n and Dempster and the spatial pster–Shafer be results show that	ng task due to olume effects a e uncertainty ca –Shafer Theory information ar elief structure. T at the proposed	the complex and noise. Se aused by these is applied to e used as feat The novelty as algorithm, ca	anatomical gmentation e effects. In brain MRI ures. In the spect of this lled FDSIS			

has satisfactory outputs on both simulated and real brain MRI datasets

Table 8 (continued)

Work	FCM	СТ	SC	PNS	SD	DT	L	VSM		
[48]	Yes	Yes	Yes	NM	Yes	Yes	EL	NM		
Model/Method in [48]	This survey aims to add some additional features for improving the classification method									
Summary of the work in [48]	Sentiment classification aims to detect information such as opinions, explicit, implicit feelings expressed in text. Most of the existing approaches are able to detect either explicit expressions or implicit expressions of sentiments in the text separately. In this proposed framework, it will detect both implicit and explicit expressions available in the meeting transcripts. It will classify the positive, negative and neutral words and also identifies the topic of the particular meeting transcripts by using fuzzy logic. This work aims to add some additional features for improving the classification method. The quality of the sentiment classification is improved using proposed fuzzy logic framework. This fuzzy logic includes features like fuzzy rules and fuzzy c-means algorithm. The quality of the output is evaluated using parameters such as precision, recall, f-measure. Here, Fuzzy C-means Clustering technique is measured in terms of purity and entropy									
[49]	Yes	Yes	Yes	NM	Yes	Yes	Yes	Yes		
Model/Method in [49]	Clustering do	ocuments wit	h labeled and u	nlabeled docum	nents using fuzz	y semi-Kmeans				
Summary of the work in [49]	While focusing on document clustering, this work presents a fuzzy semi-supervised clustering algorithm called fuzzy semi-Kmeans. The fuzzy semi-Kmeans is an extension of K-means clustering model, and it is inspired by an EM algorithm and a Gaussian mixture model. Additionally, the fuzzy semi-Kmeans provides the flexibility to employ different fuzzy membership functions to measure the distance between data. This work employs Gaussian weighting function to conduct experiments, but cosine similarity function can be used as well. This work conducts experiments on three data sets and compares fuzzy semi-Kmeans with several methods. The experimental results indicate that fuzzy semi-Kmeans can generally outperform the other methods.									
[50]	Yes	Yes	Yes	NM	NM	NM	NM	NM		
Model/Method in [50]	Aspect term extraction for sentiment analysis in large movie reviews using Gini Index feature selection method and SVM classifier									
Summary of the work in [50]	With the rapid development of the World Wide Web, electronic word-of-mouth interaction has made consumers active participants. Nowadays, a large number of reviews posted by the consumers on the Web provide valuable information to other consumers. Such information is highly essential for decision making and hence popular among the internet users. This information is very valuable not only for prospective consumers to make decisions, but also for businesses in predicting the success and sustainability. In this study, a Gini Index based feature selection method with Support Vector Machine (SVM) classifier is proposed for sentiment classification for large movie review data. The results show that our Gini Index method has better classification performance in terms of reduced error rate and accuracy									
Our study	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Model/Method in our study	Fuzzy C-Mea	ans algorithn	n for English se	ntiment classifi	cation in the Clo	oudera distribute	ed system			
Summary of our study	Firstly, we use the Fuzzy C-Means algorithm (FCM) to classify the English documents as having either a positive polarity, negative polarity or neutral polarity in the sequential environment									
	Then, we use the Fuzzy C-Means algorithm (FCM) to classify the English documents as having either positive polarity, negative polarity neutral polarity in the Cloudera distributed environment with the purpose of shortening the execution time									

sentiment classification

 Table 9 Comparison of the proposed model with the latest sentiment classification models (or the latest sentiment classification methods) in

 [51–56]

Studies	FCM	СТ	SC	PNS	SD	DT	L	VSM	
[51]	No	No	Yes	NM	Yes	Yes	Yes	vector	
Model/Method in [51] Summary of the work in [51]	The machine Opinion min towards entit With the rapid ing sites, it h analysis can h algorithms. T The main em for sentiment work in the for machine-lear for sentiment for sentiment	learning app ing or sentin ies such as p d growth in t as become r be categorize he authors su phasis of thi: classification billowing pha ning method analysis. The analysis and	proaches applied ment analysis is products and ser he availability a lecessary to ana ed into semantic urvey the machin s work is to disc mat the docume less: (1) feature s. The study also the authors conclu-	to sentiment a a study that a vices. It has al nd popularity of lyze and under orientation-bas he learning appi uss the research nt level. Machi extraction, (2) o discusses the s ade the work w future research	nalysis-based ap analyzes people ways been impo- of online review stand these revi- sed approaches, J roaches applied to a involved in app ne learning-base feature weightin standard free ben ith a comparativ directions in op	pplications so opinions or so ortant to know so sites, blogs, for ews. The main knowledge-base to sentiment ana olying machine ed approaches to g schemes, (3) is achmark datasets e study of some inion mining ar	sentiments fro what other pe ums, and soci approaches t id, and machi ilysis-based a learning mett o sentiment cl feature select s and evaluati e state-of-the- nd sentiment	om the text eople think. ial network- o sentiment ine-learning ipplications. hods mostly lassification tion, and (4) ion methods art methods analysis	
[52]	No	No	Yes	NM	Yes	Yes	NM	NM	
Model/Method in [52]	Two types of analysis, viz.	techniques , (i) corpus b	have been used ased and (ii) did	in the literature tionary or lexio	e for semantic o con or knowledg	rientation-based e based	l approach fo	»r sentiment	
Summary of the work in [52]	Two types of techniques have been used in the literature for semantic orientation-based approach for sentiment analysis, viz., (i) corpus based and (ii) dictionary or lexicon or knowledge based. In this work, the authors explore the corpus-based semantic orientation approach for sentiment analysis. Corpus-based semantic orientation approach for sentiment analysis. Corpus-based semantic orientation approach for sentiment analysis. Corpus-based semantic orientation approach requires large dataset to detect the polarity of the terms and therefore the sentiment of the text. The main problem with this approach is that it relies on the polarity of the terms that have appeared in the training corpus since polarity is computed for the terms that are in the corpus. This approach initially mines sentiment-bearing terms from the unstructured text and further computes the polarity of the terms. Most of the sentiment-bearing terms are multi-word features unlike bag-of-words, e.g., "good movie," "nice cinematography," "nice actors," etc. The performance of the semantic orientation-based approach has been limited in the literature due to inadequate coverage of the multi-word features								
[53]	No	No	Yes	NM	Yes	Yes	EL	NM	
Model/Method in [53]	New meta-lev derived from the distribution unsupervised	vel features e the sentime on of distance lexical-base	especially design ent distribution ces of x to their d methods	ned for the senti among the k n neighbors and	iment analysis of earest neighbors (iii) the docume	f short messages s of a given sho ent polarity of t	s such as: (i) ort test docu hese neighbo	information ment x, (ii) ors given by	
Summary of the work in [53]	In this work, the authors address the problem of automatically learning to classify the sentiment of short mes- sages/reviews by exploiting information derived from meta-level features i.e., features derived primarily from the original bag-of-words representation. The authors propose new meta-level features especially designed for the sentiment analysis of short messages such as: (i) information derived from the sentiment distribution among the k nearest neighbors of a given short test document x, (ii) the distribution of distances of x to their neighbors and (iii) the document polarity of these neighbors given by unsupervised lexical-based methods. The authors' approach is also capable of exploiting information from the neighborhood of document x regarding (highly noisy) data obtained from 1.6 million Twitter messages with emoticons. The set of proposed features is capable of transform- ing the original feature space into a new one, potentially smaller and more informed. Experiments performed with a substantial number of datasets (nineteen) demonstrate that the effectiveness of the proposed sentiment-based meta-level features is not only superior to the traditional bag-of-word representation (by up to 16 %) but is also superior in most cases to state-of-art meta-level features previously proposed in the literature for text classifica- tion tasks that do not take into account some idiosyncrasies of sentiment analysis. The authors' proposal is also								
[54]	No	No	Yes	NM	Yes	Yes	NM	NM	
Model/Method in [54]	Rule based m	achine learn	ing algorithms						
Summary of the work in [54]	Sentiment an ing sites etc. sentiment ana authors use S tive, weak-po on online bo accuracy of 9 Recall, and T based machin	alysis is beco where peop dysis and op entiWordNe ssitive, neutro oks and poli 07.4 % and 1 P-Rate depion ne learning a	oming a promisi le exhibit their v inion mining of t which generate al, weak-negativ tical reviews an lower error rate ets a higher effic lgorithms have	ng topic with th views on variou Web reviews us es score count v e, negative and d demonstrates . The weighted iency rate and b been performe	the strengthening the strengthening stopics. In this sing various rule- words into one o strong-negative s efficacy throug average of diff lower FP-Rate. O d through a ten-	of social media work, the focu- based machine f seven categori words. The pro- gh Kappa measu- erent accuracy Comparative exp fold cross valid	t such as blog s is to perfor learning algo es, strong-po posed approx ures, which h measures like periments on lation trainin	is, network- im effective prithms. The sitive, posi- ach is tested has a higher e Precision, various rule g model for	

Table 9(continued)

Studios	ECM	СТ	SC	DNG	۶D	DT	T	VSM		
Studies	FUM	CI	30	FN3	3D	DI	L	v 51v1		
[55]	No	No	Yes	No	No	No	EL	No		
Model/Method in [55]	A combination of the term-counting method and enhanced contextual valence shifters method									
Summary of the work in [55]	The authors explored different methods of improving the accuracy of sentiment classification. The sentiment orientation of a document can be positive $(+)$, negative $(-)$, or neutral (0). The authors combine five dictionaries into a new one with 21,137 entries. The new dictionary has many verbs, adverbs, phrases and idioms, which are not in the five previous before. The study shows that the authors' proposed method based on the combination of the term-counting method and the enhanced contextual valence shifters method has improved the accuracy of sentiment classification. The combined method has an accuracy of 68.984 % of the testing dataset, and 69.224 % on the training data set. All these methods are implemented to classify the reviews based on our new dictionary and the Internet Movie data set									
[56]	No	No	Yes	No	No	No	EL	No		
Model/Method in [56]	Naive Bayes model with N-GRAM method, Negation Handling method, Chi-Square method and Good-Turing Discounting, etc									
Summary of the work in [56]	The authors explored the Naive Bayes model with N-GRAM method, Negation Handling method, Chi-Square method and Good-Turing Discounting by selecting different thresholds of Good-Turing Discounting method and different minimum frequencies of Chi-Square method to improve the accuracy of sentiment classification									
Our work	Yes	Yes	Yes	Yes	Yes	Yes	EL	Yes		
Model/Method in our work	Fuzzy C-Means algorithm for English sentiment classification in the Cloudera distributed system									
Summary of our work	Firstly, we use the Fuzzy C-Means algorithm (FCM) to classify the English documents as having either positive polarity, negative polarity or neutral polarity in the sequential environment									
	Then, we use the Fuzzy C-Means algorithm (FCM) to classify the English documents as having either posi- tive polarity, negative polarity or neutral polarity in the Cloudera distributed environment with the purpose of shortening the execution time									

References

- Large movie review dataset (2016) http://ai.stanford.edu/~amaas/ data/sentiment/
- Singh VK, Singh VK (2015) Vector space model: an information retrieval system. International Journal of Advanced Engineering Research and Studies
- Carrera-Trejo V, Sidorov G, Miranda-Jiménez S, Moreno Ibarra M, Cadena Martínez R (2015) Latent Dirichlet allocation complement in the vector space model for multi-label text classification. International Journal of Combinatorial Optimization Problems and Informatics 6(1):7–19
- Soucy P, Mineau GW (2005) Beyond TFIDF weighting for text categorization in the vector space model. In: Proceedings of the 19th international joint conference on Artificial intelligence, USA, pp 1130–1135
- 5. Hadoop (2016). http://hadoop.apache.org
- 6. Apache (2016). http://apache.org
- 7. Cloudera (2016). http://www.cloudera.com
- Ghaffari M, Ghadiri N (2016) Ambiguity-driven fuzzy C-means clustering: how to detect uncertain clustered records. Applied Intelligence (APIN):1–12
- RJ Hathaway JC, Bezdek YHu (2000) Generalized fuzzy c-means clustering strategies using L/sub p/ norm distances. IEEE Trans Fuzzy Syst 8(5):576–582
- Tsao EC-K, Bezdek JC, Pal NR (1994) Fuzzy Kohonen clustering networks. Pattern Recogn 27(5):757–764
- Hathaway RJ, Bezdek JC (2001) Fuzzy c-means clustering of incomplete data. IEEE Trans Syst Man Cybern B (Cybern) 31(5):735–744
- Lim YW, Lee SU (1990) On the color image segmentation algorithm based on the thresholding and the fuzzy c-means techniques. Pattern Recogn 23(9):935–952

- Bezdek JC, Ehrlich R, Full W (1984) FCM: the fuzzy c-means clustering algorithm. Comput Geosci 10(2–3):191–203
- Pal NR, Bezdek JC (2002) On cluster validity for the fuzzy cmeans model. IEEE Trans Fuzzy Syst 3(3):370–379
- Pal NR, Pal K, Keller JM, Bezdek JC (2005) A possibilistic fuzzy c-means clustering algorithm. IEEE Trans Fuzzy Syst 13(4):517– 530
- Ahmed MN, Yamany SM, Mohamed N, Farag AA (2002) A modified fuzzy c-means algorithm for bias field estimation and segmentation of MRI data. IEEE Trans Med Imaging 21(3):193– 199
- Cannon RL, Dave JV, Bezdek JC (2009) Efficient implementation of the fuzzy c-means clustering algorithms. IEEE Trans Pattern Anal Mach Intell 8(2):248–255
- Bezdek JC, Hathaway RJ, Sabin MJ, Tucker WT (1987) Convergence theory for fuzzy c-means: Counterexamples and repairs. IEEE Trans Syst Man Cybern 17(5):873–877
- Hathaway RJ, Bezdek JC (1994) Nerf c-means: non-euclidean relational fuzzy clustering. Pattern Recogn 27(3):429–437
- D-Q Zhang S-C, Chen A (2004) Novel kernelized fuzzy C-means algorithm with application in medical image segmentation. Artif Intell Med 32(1):37–50
- Hathaway RJ, Davenport JW, Bezdek JC (1989) Relational duals of the c-means clustering algorithms. Pattern Recogn 22(2):205– 212
- Chuang K-S, Tzeng H-L, Chena S, Wu J, Chen T-J (2006) Fuzzy c-means clustering with spatial information for image segmentation. Comput Med Imaging Graph 30(1):9–15
- Bahrampour S, Moshiri B, Salahshoor K (2011) Weighted and constrained possibilistic C-means clustering for online fault detection and isolation. Appl Intell (APIN) 35(2):269–284
- Zhang D-Q, Chen S-C (2003) Clustering incomplete data using kernel-based fuzzy c-means algorithm. Neural Process Lett 18(3):155–162

- Hall LO, Bensaid AM, Clarke LP, Velthuizen RP (2002) A comparison of neural network and fuzzy clustering techniques in segmenting magnetic resonance images of the brain. IEEE Trans Neural Netw 3(5):672–682
- Kuo RJ, Ho LM, Hu CM (2002) Integration of self-organizing feature map and K-means algorithm for market segmentation. Comput Oper Res 29(11):1475–1493
- Kwok T, Smith K, Lozano S, Taniar D (2002) Parallel Fuzzy c-Means Clustering for Large Data Sets, Euro-Par 2002 Parallel Processing, Volume 2400 of the series Lecture Notes in Computer Science, pp 365–374
- Xylogiannopoulos KF, Karampelas P, Alhajj R (2016) Repeated patterns detection in big data using classification and parallelism on LERP Reduced Suffix Arrays. Appl Intell (APIN):1–31
- 29. Carns PH, Ligon III WB, Ross RB, Thakur R (2000) PVFS: A parallel file system for linux clusters. In: Proceedings of the extreme linux track: 4th annual linux showcase and conference
- 30. Moyer SA, Sunderam VS (1994) PIOUS: a scalable parallel I/o system for distributed computing environments. In: Proceedings of the scalable high-performance computing conference
- Shirazi BA, Kavi KM, Hurson AR (1995) Scheduling and load balancing in parallel and distributed systems, scheduling and load balancing in parallel and distributed systems, USA
- Andrews GR (1999) Foundations of parallel and distributed programming. In: Foundations of parallel and distributed programming 1st, USA
- Gropp W, Lusk E, Doss N, Skjellum A (1996) A highperformance, portable implementation of the MPI message passing interface standard. Parallel Comput 22(6):789–828
- 34. Yu Y, Isard M, Fetterly D, Budiu M, Erlingsson Ú, Gunda PK, Currey J (2008) dryadLINQ: a system for general-purpose distributed data-parallel computing using a high-level language symposium on operating system design and implementation (OSDI)
- 35. Shvachko K, Kuang H, Radia S, Chansler R (2010) The hadoop distributed file system
- 36. Guerrero JM, Matas J, Garcia de Vicuna L, Castilla M, Miret J (2007) Decentralized control for parallel operation of distributed generation inverters using resistive output impedance. IEEE Trans Ind Electron 54:2
- van Steen M, Homburg P, Tanenbaum AS (1999) Globe: a widearea distributed system. IEEE Concurr 7(1):70–78
- Shende SS, Malony AD (2006) The tau parallel performance system. Int J High Perform Comput Appl 20(2):287–311
- Bagrodia R, Meyer R, Takai M, Chen Y-A, Zeng X, Martin J, Song HY (1998) Parsec: a parallel simulation environment for complex systems. Computer 31(10):77–85
- RumelHart DE, Hinton GE, McClelland JL (1986) A general framework for parallel distributed processing. In: Parallel distributed processing: explorations in the microstructure of cognition, USA, vol 1, pp 45–76
- Ikudome K, Fox GC, Kolawa A, Flower JW (1990) An automatic and symbolic parallelization system for distributed memory parallel computers. In: Proceedings of the fifth distributed memory computing conference
- 42. Wang HO, Tanaka K, Griffin M (1995) Parallel distributed compensation of nonlinear systems by Takagi-Sugeno fuzzy model
- Poria S, Gelbukh A, Cambria E, Hussain A, Huang G-B (2014) EmoSenticSpace: a novel framework for affective common-sense reasoning. Knowl-Based Syst 69:108–123
- 44. Poria S, Gelbukh A, Das D, Bandyopadhyay S (2013) Fuzzy clustering for semi-supervised learning – case study: construction of an emotion lexicon. In: Advances in artificial intelligence, volume 7629 of the series lecture notes in computer science, pp 73–86
- Vinchurkar SV, Nirkhi SM (2012) feature extraction of product from customer feedback through blog. International Journal of Emerging Technology and Advanced Engineering 2(1):2250– 2459

- 46. IndiraPriya P, Ghosh DK (2013) A Survey on Different Clustering Algorithms in Data Mining Technique. International Journal of Modern Engineering Research (IJMER) 3(1):267–274
- Ghasemi J, Ghaderi R, Karami Mollaei MR, Hojjatoleslami SA (2013) A novel fuzzy Dempster–Shafer inference system for brain MRI segmentation. Inf Sci 223:205–220
- Sheeba JI, Vivekanandan K (2014) A fuzzy logic based on sentiment classification. International Journal of Data Mining & Knowledge Management Process (IJDKP) 4(4)
- Liu C-L, Chang T-H, Li H-H (2013) Clustering documents with labeled and unlabeled documents using fuzzy semi-Kmeans. Fuzzy Sets Syst 221:48–64
- 50. Manek AS, Deepa Shenoy P, Chandra Mohan M, Venugopal KR (2016) Aspect term extraction for sentiment analysis in large movie reviews using gini index feature selection method and SVM classifier. World wide web, 1–20. doi:10.1007/s11280-015-0381-x. Print ISSN1386-145x, US
- Agarwal B, Mittal N (2016) Machine learning approach for sentiment analysis. Prominent feature extraction for sentiment analysis, 21–45. doi:10.1007/978-3-319-25343-5_3. Print ISBN 978-3-319-25341-1
- Agarwal B, Mittal N (2016) Semantic orientation-based approach for sentiment analysis. Prominent feature extraction for sentiment analysis, 77–88. doi:10.1007/978-3-319-25343-5_6. Print ISBN 978-3-319-25341-1
- 53. Canuto S, André M, Gonçalves FB (2016) Exploiting new sentiment-based meta-level features for effective sentiment analysis. In: Proceedings of the ninth ACM international conference on web search and data mining (WSDM '16), New York, USA, pp 53–62
- 54. Ahmed S, Danti A (2016) Effective sentimental analysis and opinion mining of web reviews using rule based classifiers. Computational Intelligence in Data Mining 1:171– 179. doi:10.1007/978-81-322-2734-2_18. Print ISBN 978-81-322-2732-8, India
- Phu VN, Tuoi PT (2014) Sentiment classification using enhanced contextual valence shifters. In: International Conference on Asian Language Processing (IALP), pp 224–229
- 56. Tran VTN, Phu VN, Tuoi PT (2014) Learning more chi square feature selection to improve the fastest and most accurate sentiment classification. In: The third asian conference on information systems (ACIS 2014)



Vo Ngoc Phu received B. Sc Degree in Computer Science from Ho Chi Minh City University of Science, Ho Chi Minh City, Viet Nam in 2000, M. Sc Degree in Computer Science from Ho Chi Minh City University of Technology, Ho Chi Minh City, Viet Nam in 2014. He is currently a lecturer and a researcher in Ton Duc Thang University, Ho Chi Minh City, Vietnam and Duy Tan University, Da Nang City, Vietnam. Nguyen Duy Dat received B. Sc Degree from Ho Chi Minh City University of Technology and Education, Ho Chi Minh City, Vietnam in 2013; M. Sc Degree from Ho Chi Minh City University of Technology, Ho Chi Minh City, Viet Nam in 2014. He is currently a lecturer in Cao Thang College, Ho Chi Minh City, Vietnam.

Vo Thi Ngoc Tran received her B.Sc Degree and M. Sc Degree in Computer Science from Ho Chi Minh City University of Technology, Vietnam National University, HoChi Minh City, Vietnam in 1989 and 1990. **Vo Thi Ngoc Chau** received her B.Sc Degree, M. Sc Degree, and Dr.Sc Degree in Computer Science from Ho Chi Minh City University of Technology, Vietnam National University, Ho Chi Minh City, Vietnam in 1989, 1990 and 1995, respectively. She is currently a lecturer at Ho Chi Minh City University of Technology, Vietnam National University, Ho Chi Minh City, Vietnam.

Tuan A. Nguyen received his B.Sc Degree in Computer Science from Ho Chi Minh City University of Science, Ho Chi Minh City, Viet Nam in 1989; M. Sc. Degree and Dr. Sc. Degree in Computer Science from La Trobe University, Australia in 1990 and in 1995, respectively. He is currently a lecturer at University of Information Technology, Vietnam National University of Hochiminh City.